Author's response to reviews

Title: Caregiver Stress In Stroke Survivors: Data from a Tertiary care Hospital: A Cross Sectional Survey

Authors:

Qurat Ul Ain (guratulain_k90@hotmail.com)
Nayab Zaheer Dar (nayabdar@hotmail.com)
Arsalan Ahmad (arsalanahmad65@gmail.com)
Saad Munzar (smunzar@hotmail.com)
Abdul Wahab Yousafzai (wahab.yousafzai@gmail.com)

Version: 5 Date: 18 October 2014

Author's response to reviews:

DEAR CHIEF EDITOR

BIOMED CENTRAL PSYCHOLOGY,

These are the following revisions made after reading reviewer's Report about my research Caregiver Stress in Stroke Survivors: Data from a Tertiary care Hospital: A Cross Sectional Survey

These includes following:

1. Second colon used “:” has been replaced from the title to a dash (-) now title is Caregiver Stress in Stroke Survivors: Data from a Tertiary care Hospital - A Cross Sectional Survey

2. Research design is added in the abstract.

3. Objective is better explained in abstract now.

4. In line 142 "insignificant" was replaced with "not statistically significant".

5. In line 155 "Spearman" is now with capital letter

6. In line 157 "(Table 3)" now before the dot.

7. References subtitle added before the references.

8. Style of references is changed.

9. In materials and methods para 3 was interchanged with para 1, from line 59 onwards.

10. We have tried covering every point of change asked by the editors.

Looking forward

Author

Qurat Ul Ain
Shifa college of medicine
Islamabad Pakistan